Author's response to reviews

Title: Maternal efficacy and sedentary behavior rules predict child obesity resilience.

Authors:

David Crawford Prof (david.crawford@deakin.edu.au)
Kylie Ball Prof (kylie.ball@deakin.edu.au)
Verity Cleland Dr (Verity.Cleland@utas.edu.au)
Lukar Thornton Dr (lukar.thornton@deakin.edu.au)
Gavin Abbott Dr (gavin.abbott@deakin.edu.au)
Sarah A McNaughton Dr (sarah.mcnaughton@deakin.edu.au)
Karen J Campbell Dr (kcamp@deakin.edu.au)
Johannes Brug Prof (j.brug@vumc.nl)
Jo Salmon Prof (jo.salmon@deakin.edu.au)
Anna Timperio Dr (anna.timperio@deakin.edu.au)

Version: 2
Date: 2 February 2015

Author's response to reviews:

Dear Editor

Re: Maternal efficacy and sedentary behaviour rules predict child obesity resilience

We advise that the following changes have been made to the manuscript to enable the peer review process to commence.

Line and page numbers have been included in the manuscript.
the abstract has been formatted according to the guidelines.
Author contributions have been included in the manuscript and the competing interest section has been moved.

Thank you for your consideration

David